Author’s response to reviews

Title: Knowledge of Pelvic Floor Disorders in Women Seeking Primary Care: A Cross-sectional Study

Authors:

Chi Chiung Grace Chen (cchen127@jhmi.edu)
Jacob Cox (jtcox@jhmi.edu)
Chloe Yuan (yuanc@hawaii.edu)
Lauren Thomaier (lthomai1@jhmi.edu)
Sonia Dutta (sonia.dutta@gmail.com)

Version: 3 Date: 29 Apr 2019

Author’s response to reviews:

Editor Comments

1. Thank you for including information regarding your ethics approval. However, in the Ethics approval section of your Declarations, please provide the full name of the committee that approved this study and a reference number if appropriate.

   This has already been done in the previous version of the manuscript which we have highlighted in this version for your ease of review. We did add more detailed info re IRB in the methods part of the manuscript.

2. Please remove the “Study Registration” section from the Abstract, as this should only include a trial registration number if your study was registered in a clinical trials registry.

   Done.

3. Please provide a List of abbreviations after the Conclusions section. If abbreviations are used in the text, they should be defined in the text at first use and included in this list.

   Done.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should
remain uploaded as separate files.

Done.